Author's response to reviews

Title: Impact of leg lengthening on viscoelastic properties of the deep fascia

Authors:

Hai-Qiang Wang (hqwang@fmmu.edu.cn)
Yi-Yong Wei (yiyongwei@yahoo.com.cn)
Zi-Xiang Wu (wuzixiang@hotmail.com)
Zhuo-Jing Luo (ziluo@fmmu.edu.cn)

Version: 5 Date: 17 August 2009

Author's response to reviews: see over
Dear Editor,

Thank you very much for your suggestions for formatting changes on our manuscript entitled “Impact of leg lengthening on viscoelastic properties of the deep fascia” (MS 1677819625335357). Having read them carefully, we made the changes according to your suggestions. Please find the responses to your suggestions as following. Hope these will make it ready for publication.

Kindest regards.

Hai-Qiang Wang, MD, PhD

Institute of Orthopaedics, Xijing Hospital, Fourth Military Medical University, Xi’an, People’s Republic of China, 710032

Encl.
Corrections and changes according to the suggestions by the editor

Formatting changes requested

-----------------------------

Major revisions (We require the author to make these changes)
Conclusions - please add a Conclusions section following the Discussion.

We have added a conclusions section following the Discussion.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above)
Highlighted text - please remove this from the manuscript. You may replace it with bold if you wish.
We have removed the highlighted yellow colour from the manuscript.

Figure legends - please move to a position after the references.
We have moved them to the position after the references.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.
We checked our manuscript carefully and guaranteed that there were no typographical errors according to our knowledge.